Author's response to reviews

Title: A child presenting with tuberculous spondylitis in a single cervical (C3) vertebra: a case report

Authors:

Manouri Senanayake Prof. (manouri.senanayake@gmail.com)
Irantha Karunaratne (iranthakarunaratne@hotmail.com)

Version: 2 Date: 27 March 2014

Author's response to reviews:

Dear Editors,

We have revised as suggested, the following.

1. Title has been changed.
3. Keywords included after Abstract section.
4. 'Female ' substituted for "girl"
5. Abbreviations, Conflict of interest, Authors contributions, Patients consent, have been included before References.
6. Title page and Abstract have been placed as Page 1 and 2

Looking forward to an early review,
Sincerely,
Prof Manouri Senanayake